Author's response to reviews

Title: Delayed union of femoral fractures in older rats: decreased gene expression

Authors:

Ralph A. Meyer Jr. (rmeyer@carolinas.org)
Martha H. Meyer (mmeyer@carolinas.org)
Laura S. Phieffer (lphieffer@carolinas.org)
David M. Banks (dbanks@carolinas.org)

Version: 4 Date: 28 Jun 2001

Carole Mongin,

Thank you for the review of our manuscript. I have made the minor changes that were requested:

1. The abstract was shortened to 250 words
2. Materials and Methods were moved to follow the Background section.
3. The title was written without caps.
4. The conflict of interest section was appended to the end of the acknowledgements.

Please let us know if any further changes are needed.

Thanks for your help.

Ralph Meyer